Notes of the RICH1 Phone meeting 28/04/2006
People Present:
Bristol:  Richard(part)
CERN:  Carmelo, Christophe, Fabio, Olav 
IC:         Bill, Dave, Gabriel, Richard, Trevor
Oxford: Neville
RAL:     Peter, Derek
Apologies for absence:
Bristol: Nick 
Oxford: Rohan
RAL: Mike
Discussed Agenda Topics:
1. Gas Enclosure 
Plan to bring the GE to the pit around May 20th. The date is not yet confirmed. The latest official installation date is 8th June06. 

Mounts:  There were small discrepancies between the drawings and the manufacturing of the GE, but nothing serious. The pieces have to be returned to London for rectification. 
Painting: A second set of aerogel tests will begin next week; the first test with polyurethane paint from Christophe, and a second test with the RICH2 paint. The previous test with the RICH2 paint gave no difference in clarity but a small change in n (2( effect).  The delivery time for the RICH2 paint is ~1-2 weeks and ~2-3 weeks for polyurethane. Decided to go ahead and order the RICH2 paint (action Christophe).
Survey:  The GE has been taken apart. Next Tuesday morning it will be put on the trolley and be prepared for the survey (action Dave, Fabio). The side door survey bolts will be brought to CERN by Dave (Tuesday).  The survey will be performed outside the clean room in Bat 156 and it will take ~3 days. The surveyors will start preparing things on Tuesday.
Installation pieces:  All has been shipped to CERN. The foreseen arrival date is May 10th. Waiting from Rolf a date for the GE installation safety review.
Aerogel rail test fitting: Tuesday 9th May (action Clara).
2. VELO Seal
It is packed up in its crate and it is ready to be transported to CERN (action RAL).  The installation tooling drawings are completed and are ready for quotation, but they need final approval for manufacture. The installation material (table etc.) will be delivered to the IC workshop next week.
3. Exit Window
The EW is ready. Some diaphragm samples will be prepared with 50% and 75% pigment respectively for the opacity tests. The mould shape is still under development. A new mould will be manufactured for a new shape and the drawings are going out for quotation next week. It is hoped that with a new mould shape the leak rate can be reduced.  The leak is mainly from the outer diameter but there is no quantitative estimate. An installation plan needs to be developed soon and this will require a visit to IP8 by Mike/Bill/Rolf (action Mike/Bill). The beampipe will be inserted ~ 7 August06.

4. Spherical Mirrors
The samples from CMA have been received. Two spherical mirrors 15cm diameter and 3 flat samples.  Two flat samples will be used for the mechanical tests and one flat sample will be given to CERN safety for the fire test (action Fabio). 
The spherical mirrors were measured: 
Sample 1: D0=0.97mm, R=1876mm

Sample 2: D0=0.72mm, R=1882mm.

Andre is setting up the reflectometer, ~10-14 days to be ready. Help has been offered to speed up things. Kenneth has helped out with the testing of the photomultipliers. Two flat samples have been brought to metrology to measure overall dimensions, weight and surface profile; then they will be brought to the mechanics lab for the mechanical tests. The CMA demonstration mirror is being built. It will be in CERN ~3 weeks. The blank for the mandrel cracked during the cool down. This gives a four week delay. The mandrel should be in CMA by the start of July06. Geoff and Trevor will visit CMA 8-9 May06.  

5. Flat mirrors
Five flat mirrors have been coated. There is no news from Milan. He should bring the remaining ~5 mirrors in ~1-2 weeks.  All is proceeding fine.
Retainer Ring:  The first gluing test on the dummy mirror looks good. If the gluing is satisfactory the retainer rings will be ordered (action Christophe). The gluing will start ~3 weeks.
Mechanics:  The support mechanics drawings are to be prepared by Richard (action Richard).
6. HPD Assembly
The green light for the 89.5 mm pitch has been given. Rohan’s second iteration for the HPD rail system is converging to the requirements and he will meet with the IC team next week. Supra36 shields (1mm thick) will be ordered (action Dave/Tito). Material for the ladder system mechanics has been ordered.

7.  Quartz Windows
The final dimensions have been released by Trevor.  Mike will make the order (action Mike).

8.  LAMS

A radiation tolerant CID camera has been delivered and is being tested.  Need to check with CERN safety if PVC cables can be used.
9.  Magnetic Distortions

A dedicated meeting will be held this afternoon with Syracuse.  The Syracuse system is difficult to accommodate in the photon funnel, but it is still the baseline. Although Syracuse rejected the projection option, IC is conducting tests using a projector system to check its feasibility as a backup solution.  Four small projectors could illuminate all HPDs, though with shadowing loss (to be evaluated).

10. Next meeting
Friday 12th of May 2006 at 10:00(GMT) 11:00(CET). 
